CENWP-OP-B                                                                                                                                  17 April 2006

MEMORANDUM FOR CENWP-OP, ATTN: Randy Bailey

SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for Week 15 of 2006, which covers the period from 09 April through 15 April 2006.
1 OPERATION SUMMARY

1.1 Daily average river flows ranged from 268.6 to 348.6 kcfs.  Daily average project tailwater elevation ranged from 21.1’ to 24.9’ msl.  Daily average powerhouse forebay elevation ranged from 73.5’ to 75.9 msl.  Daily average river temperature ranged from 48 to 49(F.  Secchi disk measurements averaged 2.5 to 6.0 feet.

1.2 Spill Bays 1 and 18, which supply attraction flow for Cascades Island and B-Branch respectively, are included in the spill pattern.  Daily average spill ranged from 63.5 to 143.2 kcfs.
1.3 Unit Operation.

1.3.1 Unit 0 remains out of service due to governor trouble.

1.3.2 Unit 2 remains out of service for turbine rehab.
1.3.3 Unit 10 remains out of service for turbine rehab.

1.3.4 Unit 11 remains out of service due to a bearing problem.

1.3.5 Unit 12 remains out of service for VBS modifications.
1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore fish ladder.  The units are placed on standby, as necessary, during nighttime hours to float debris.

1.4.1 Fish Unit 1 and 2 were run at reduced load while removing floating orifice gates for modifications from 0830 on 10 April until 1612 on 13 April.  Load was reduced only to allow for removal and fishways were returned to operating criteria once each gate was removed from the water.  See 6.1.5
1.5 Gatewell drawdowns are measured on Wednesdays.  VBS drawdowns are measured daily on Units 17 and 13.

2 ACTIVITIES

2.1 AWS Closures.  The following AWS valves were closed this reporting week:  
2.1.1 FV3-9 closed for cleaning from 1625 until 1635 on 12 April.   
2.2 STS/VBS Inspections.   No STSs/VBSs were inspected this reporting week.
2.3 Dewatering and Fish Salvage.  The following dewatering or salvage activities occurred this reporting week:
2.3.1 Unit 12 tail logs - 16 sturgeon, two catfish, one juvenile chinook, and one sculpin were recovered.  Five of the sturgeon were dead on arrival, one of which was killed by contact with the tail log picking beam.  All live fish were released to the tailrace at the WA shore boat ramp.  
3 RESEARCH
3.1 ERDC TDG.  Sensors remain installed in the tailrace.
3.2 Fish Field Unit.  Sea lion observation and acoustic deterrent evaluations continue.

3.3 NOAA Fisheries lamprey.  Permanent installation of the BI AWS lamprey ramp completed.
3.4 ODFW. Sampling Northern Pikeminnow in the BRZ this reporting week.  
3.5 PSMFC.  10 adult steelhead and one salmon were observed crossing the separator bars this week. 
3.6 U of I.  Radio tagging of spring chinook for SLED evaluation continues in the AFF.

3.7 USDA Wildlife Services.  Sea lion hazing continues using 15mm cracker/screamer shells and rubber bullets.

3.8 WDFW Fish Counting.  Visual fish counting continues.
3.9 WDFW/ODFW.  Sea lion hazing from boats continues.

4 FISHWAYS 

4.1 Project biologists inspected 09 – 15 April.  NOAA Fisheries inspected on 14 April
4.2  Fish Passage Plan Observations.
	DATE
	LOCATion
	FPP Deviation
	CAUSE
	RESPONSE
	TR#

	All week
	PH2 SCCH
	Velocity <1.5 fps
	Improper installation
	Will repair in 06/07
	

	All week
	Cascades Is.
	FG6-12 closed
	Bent shaft
	Will repair in 06/07
	

	All week
	PH1 CCH Diffusers
	FG2-22A closed, 2-19 stuck half open
	Unknown.  Failed during water up.
	Will repair in 07/08
	

	4/10-4/15
	PH2 SUE
	Entrance gate < 13’
	Unknown
	TR submitted for calibration
	06-29824

	4/9-4/10
	Casc. Is. entrance
	Differential low (0.8’)
	Unknown
	Operator adjusted.
	


Table 2.  FPP Out of Criteria items.

4.3 Fish Ladder Temperatures.  Measurements are taken three weirs down from the count station at Bradford Island and at the junction of the UMT and the main ladder at Washington shore.  
	Date
	Wash. Shore
	Bradford Is.

	4/9
	48.5
	48.6

	4/10
	48.5
	48.7

	4/11
	48.4
	48.6

	4/12
	48.4
	48.6

	4/13
	48.7
	48.8

	4/14
	48.9
	48.9

	4/15
	48.9
	48.9


4.4 Sensor Calibration.  Calibration was checked on 15 April.  
4.5 Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week.  

5 CONSTRUCTION
5.1 B1 ITS Wall Removal.  A letter report is expected in 2006.

5.2 B2CC PIT Tag Detector.  Construction complete.  Opened for the season at 0815 on 13 April.
5.3 B2 VBS modifications.  VBS delivery occurred on 3 April.  
5.4 PH1 STS/VBS removal and TIES removal.  Six TIES and 23 PH1 STSs/VBSs will be scrapped.

5.5 SMF Full flow PIT Tag Detectors.  Construction continues.
6 FISHWAY STATUS

6.1  Adult Fishways.    
6.1.1 Bradford Island and Cascades Island PLC’s are in manual.  
6.1.2 New sensors for B-branch and Cascades Island tailrace, entrance and ladder have been installed. 
6.1.3 Cascades Island diffuser FG6-12 has a bent shaft and remains out of service until winter 2006/07. 
6.1.4 SLADDs and SLEDs are installed at all fishway entrances. 
6.1.5 All floating orifice gates at PH2 have been modified and installed.  See 1.4.1 
6.2 Juvenile Fishways.
6.2.1 DSM1 remains out of service until 15 September.
6.2.2 The B2CC was open from 1018 on 10 April until 1035 on 11 April for PIT antenna work.  It was opened at 0815 on 13 April and remains open.  
6.2.3 DSM2 ERG tripped open at 0940 until 0955 on 12 April due to high water level.  Air burst system PLC not controlling properly in auto.  DSM2 mechanical screen cleaners are in auto on six hour cycles until the airburst system is repaired.
7 HAZMAT, SPILLS AND CLEANUP - Nothing to report.
8 TELETYPES AND OTHER NOTEWORTHY ITEMS. 
8.1 A teletype issued on 13 April authorized the project to begin operation of the corner collector at 0815 as outlined in the Fish Passage Plan.

JAMES R. MAHAR, P.E.

Operations Manager
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